

The use of contingent valuation in assessing marine and coastal ecosystems' water quality: A review

Halkos, George

University of Thessaly, Department of Economics

October 2012

Online at https://mpra.ub.uni-muenchen.de/42183/ MPRA Paper No. 42183, posted 25 Oct 2012 09:16 UTC

The use of contingent valuation in assessing marine and coastal ecosystems' water quality: A review

By

George E. Halkos and Georgia K. Galani

Laboratory of Operations Research, Department of Economics,

University of Thessaly, Korai 43, 38333, Volos, Greece

Abstract

Marine and coastal ecosystems are of high importance owing to the mankind dependence on the goods and services provided. As water quality is one perspective of healthy marine and coastal ecosystems, the aim of this study is to review as more as possible surveys conducted worldwide and in Greece. Due to the lack of an official market to valuate non-marketed goods and services, contingent valuation is applied intensively in order to provide the policy makers and the society generally with the specific values derived by a developed hypothetical market. In addition, this study reviews the crucial but debatable notion of value, the theoretical framework in accordance with the existing statistical models to estimate the welfare measures and the numerous disadvantages that have to be taken into account in order to implement a reliable contingent valuation survey.

Keywords: Contingent valuation; willingness to pay; willingness to accept; marine and coastal ecosystems.

JEL codes: Q51; Q57; Q25; Q28; C80; D12.

1. Introduction

Marine and coastal ecosystems play an important role in the balance of the environment as they interrelate and interact dynamically. The coverage of water amounts to more than 70% and the remainder consists of land area (Burke *et al.*, 2001; UNEP, 2006). The total length of the world coastline extends over 350,000-1,000,000 km and circa 84% of the countries of the world that have a coastline within its extent display a variety of geomorphological types and ecosystems (Martinez, *et al.*, 2007). Moreover, it is indicative the fact that human population lives within 100 km of the coast (Cohen *et al.*, 1997; Gommes *et al.*, 1998; Burke *et al.*, 2001).

Mankind is strongly dependent on marine and coastal ecosystems and it is attracted by the environmental goods and services that are plentiful in these types of ecosystems and that influence their choice to live permanently for leisure, recreation, and tourism or even for commercial reasons (Martinez, *et al.*, 2007). People's well being, basic needs and economy rely upon the exploitation of the most benefits provided by the ecosystems. According to the Millennium Ecosystem Assessment which is an international initiative that started in 2001, ecosystem services can be further divided into four categories. Particularly, the category of *provisioning services* like food and water, the category of *regulating services* such as the regulation of climate, wastes and water quality, the *cultural services* like recreational activities and *aesthetic and supporting services* referring to the wide range of habitats that serve in life cycle, productivity and commerce.

On the other hand, humans pose a serious threat to those ecosystems resulting in degradation. The biodiversity loss, the poor water quality and the sea level rise provoked by anthropogenic pressures are some of the challenges that marine and coastal ecosystems has to encounter without profoundly depicting the integrated imagine of the disaster (Halpern, 2007, 2008). Salm *et al.* (2000) reported that the environmental degradation of marine and coastal ecosystems is multifaceted in term of the various human constructions which aim to contribute to increasing profits but in essence replace the natural environment with harbors, industries, dams and settlements.

Lack of knowledge concerning the goods and services that ecosystems provide may lead to inefficient policy to manage the adverse consequences of the mankind activity. To date, direct markets attained to capture the value of some fundamental for the survival goods but failed to capture others such as recreational, cultural or aesthetic services named as non-market values (Pendleton, *et al.*, 2007). A rising area of study is how to value the non-marketed goods and services provided by natural ecosystems. We have to understand though the important notion of value from the ecological and economic perspective.

There is a serious debate between the two separate frameworks of ecology and economy. Both are characterized by complexity of the dynamic processes. As regards the economic perspective, the environmental ecosystem functions are evaluated concerning their importance for human welfare. According to the ecologists, human welfare notion is very restricted as the economic system evolves within the dynamics of a larger system, the environmental. So, it is obvious for the ecologists that the economic value of ecosystem is not representative of the real value, as the assessment of ecosystems depends on the hypothesis of the stability and the inaction of the environmental functions (Limburg *et al.*, 2002).

The necessity of the collaboration of the two different frameworks to determine the value of nature has been an issue of a long-term debate of the scientists. Winter and Lockwood (2005) developed a model in order to incorporate different

value types such as intrinsic and instrumental to estimate the impact on the future preferences for the natural environment.

Due to market imperfections or missing markets mainly for non-use values it is very important to conduct valuation studies. Economists with the aim to estimate monetary values of environmental goods and services, employ methods such as the direct market valuation, the revealed preference methods and methods of the stated preference. In our study we will review the existing academic work concerning the non-tradable goods and services of marine and coastal ecosystems with an emphasis to water quality. The review includes the contingent valuation approach which is a category of the stated preference methods used to estimate the total benefits including both the use and non-use values (Eom & Larson, 2006).

The methods of the stated preferences are extensively used in the economic environment over the past two decades (Carson, 2000). One of the key features of the methods is that they are applied in the valuation of non-use values. The basic assumption of these methods is that they rely primarily on research through questionnaires in which preferences are elicited by the participants in the research. The questions are ranked in a manner so as to reveal directly and indirectly the monetary value of the resource under consideration. Direct questions have the form of "How much are you willing to pay?" Or "Are you willing to pay an amount X?". At the same time, to elicit respondents' preferences according to the indirect approach includes options with different features that participants are asked to choose (Pearce, 2002). In fact, the absence of a formal market is substituted by a hypothetical where values are estimated through the willingness to pay or to accept of the 'consumers'. Two types of stated preference methods are the contingent valuation method and the choice modelling. Furthermore, it is important to comment on issues of the economic theory that constitute the basis for the formulation of the explicit goals of the non-market valuation. Environmental economics inspired by the standard neoclassical price theory have developed the theoretical framework of non-market valuation. The fundamental principle of neoclassical economic theory is the individuals' preferences' for goods and the element of preference ordering. Willingness to pay or to accept (WTP/WTA) welfare measures can be derived from a statistical model that it is based on different perspectives such as parametric, non-parametric and semi-nonparametric.

Although the approach of contingent valuation is dominant in environmental valuation there are controversies concerning various problems that have to be dealt with such as the information effect, strategic behaviour, elicitation format, embedding effect, and hypothetical bias and protest responses. Taking into account the problems of the contingent valuation approach is critical in the conduction of an efficient estimation of the welfare measures stated by the participants in a survey.

The paper is organized as follows. Section 2 discusses the notion of environmental value that is being a debatable issue between economists and ecologists. Section 3 presents a literature review including non-market valuation of water quality in marine and coastal ecosystems. Section 4 describes important components of the economic theory which is the basis for the calculation of the welfare estimates. Furthermore, section 5 provides a critique as regards to different problems of contingent valuation approach and while the last section concludes the paper.

2. The notion of value: The debate between economists and ecologists

According to Costanza (2000), the term of value is used to denote the contribution of an item to achieve an objective or a goal. However, the term of value differs substantially in the framework of different disciplines such as Economics and Ecology. It is of crucial importance to shed light into different attempts and theories to interpret the 'value' (Goulder *et al.*, 1997). 'Economism Theory' and 'Intrinsic Value Theory' compete with each other as two polar opposite theories concerning the interpretation of value from the side of economists and ecologists (Norton, 2000).

There is the anthropocentric approach that usually economists tend to support. In accordance with this approach the value of a specific good and service is based on the ability to contribute to human well-being/utility directly and indirectly (Bockstael, 2000; Farber *et al.*, 2002; Goulder & Kennedy, 2010). Instrumental values that are basically anthropocentric can be further categorized in use or non-use values. This distinction depends on whether or not they are traded in formal markets. Use values include the benefits derived from the environment directly and indirectly. For example, direct use values of marine and coastal ecosystems include the extractive (food, fish, wood, medicines, etc.) and non-extractive (aesthetic, recreational, tourism, etc.) uses tradable in formal markets.

On the other hand, non-use values can be classified to bequest, existence and option values (Turner *et al.*, 2000; Winter, 2007). Bequest value relates to an altruistic motive as far as intergenerational equity (Cicchetti & Wilde, 1992; Loomis, 1988). Existence value is associated with the satisfaction derived, relying on the knowledge that a natural resource or environmental good exists (Kruttilla, 1967; Cicchetti & Wilde, 1992). Furthermore, the option value refers to the benefit derived from the

knowledge that a resource can be utilized in the future (Cicchetti & Wilde, 1992; Kruttilla, 1967; Arrow & Fisher, 1974; Winter, 2007; Turner *et al.*, 2000).

However, there is a strongly associated notion which is often mentioned in the literature with the option value that is quasi-option value. As supported by Arrow & Fisher (1974) and Henry (1974a, b) quasi-option value refers to an extra degree of uncertainty for the preservation of the natural resources used by the future generations. Finally, the separate segments of the instrumental value can be combined to form the total economic value of natural resources.

On the other hand, the interpretation of value can be different according to the biocentric approach in the sense that, in contrast to the anthropocentric approach, natural resources values can be estimated without any reference to human's needs satisfaction (Farber *et al.*, 2002; Norton, 2000; Winter, 2007; Goulder & Kennedy, 2010). As stated by O'Neill (1992), intrinsic value has an opposite meaning with the instrumental value. Specifically the intrinsic value of an object or action can be assessed by the contribution to the preservation of the health and coherence of ecosystems and species by themselves (Farber *et al.*, 2002).

In line with Vilkka (1997) statement, nature has intrinsic value regardless of human well-being. Even though in biocentric approach it is better to avoid using the notion of value as in the nature does not exist a value system, or a goal to be achieved, sometimes ecologists use the concept of value in a common way as economists. For example, some of basic concepts in biology are the evolution and co-evolution. As to the evolution by natural selection, one of the basic principles of biology, there is not an evident goal to be achieved; however there is still the goal of the 'survival'. Moreover, as regards to the importance of co-evolution or the interrelation of ecosystems, one can state that there is the implied contribution to human survival (Farber *et al.*, 2002).

The continuation of the debate between ecologists and ecomomists is based on the complementarity or substitutability. As the nature includes both instrumental and intrinsic value as stated by Halkos and Matsiori (2011), we may refer to these different approaches as complementary parts that aim to interpret an integral system. Costanza (2006) is one of the authors that support the concept of the complementarity of the two rationales. One of the most important challenges in the valuation of different ecosystems is to combine the different perceptions of value from the side of ecologists and economists (De Groot *et al.*, 2002; Pearce *et al.*, 2006). Hence, the combination of the two perspectives is crucial for the global conservation and sustainability (Millennium Ecosystem Assessment, 2005).

Generally, marine and coastal ecosystems provide the basics for the survival such as food provision, employment via fishing sector, other marketable goods such as genetic, medical or ornamental, quality in our life via tourism and recreation and many times security from natural disasters (UNEP, 2006). In order to valuate marine and coastal ecosystems it is necessary to specify a goal or objective as for example the maintenance of quality of bathing water in the coast under discussion.

Next, it is necessary to identify the contributions of the different components and functions of the ecosystem and finally to identify a management plan in order to set the operational objectives to achieve the specified goal (Katsanevakis *et al.*, 2011). It is of major importance though, the decoding of the total economic value of marine and coastal ecosystem which is the sum of use and non-use values as it is depicted in Figure 1, in order to shed light into maximization of the social welfare.



Figure 1: The distinction between the human and non-human values

3. Non-market valuation of marine and coastal ecosystems: A literature review of contingent valuation applications

The approach of contingent valuation is applied when there is no real market. Thus, by creating a hypothetical market, it is possible to valuate the non-tradable goods or services. The formal markets cannot generate prices for public goods due to their non-excludability and non-rivalry nature. As a result, we cannot depict the value of non- tradable goods and simultaneously the individuals' preferences. As it is claimed by Ciriacy-Wantrup (1947), who was the first to propose the approach of contingent valuation, individuals are encouraged to express via interviews the magnitude of the satisfaction by using the goods under estimation (Hanemann, 1994). In other words, participants in the survey are prompted to express their maximum WTP for an improvement or alternatively their minimum willingness to accept a compensation for the loss of an environmental good or service. Nevertheless, it is difficult for people to state the value of trade-offs due to not being familiar with the hypothetical market and the lack of information. The contingent valuation approach has been expanded in many areas of economics such as health economics (O'Brien & Gafni, 1996; Ryan, 2004; Smith, 2003; Diener *et al.*, 1998; Borghi & Jan, 2008; O'Shea *et al.*, 2008; Ryan & Watson, 2008), cultural economics (Lockwood *et al.*, 1996; Noonan, 2002; Kim *et al.*, 2007 Báez, 2009; Herrero *et al.*, 2011), transportation (Md Nor *et al.*, 2003), marketing (Louviere & Woodworth, 1983) and in the field of environmental economics (Hanemann, 1994; Boxal *et al.*, 1996; Adamowicz *et al.*, 1998; Hanley *et al.*, 1998; Christie & Azevedo, 2002; Alias *et al.*, 2008; Skourtos *et al.*, 2005; Remoundou *et al.*, 2009; Halkos & Jones, 2012).

Valuing marine and coastal ecosystems is among the most demanding issues in Environmental Economics. The estimation of the value of marine and coastal ecosystems is based on an instrumental viewpoint, whereby the value is considered as the interaction between the individuals and the flows of ecosystem goods and services. Additionally, the anthropocentric perspective of marine and coastal ecosystems implies the utility derived from human beings and the total impact in human welfare (Nunes & van den Bergh, 2001; Nunes *et al.*, 2009).

Fisher *et al.* (2009) claim that ecosystem goods and services are interrelated in terms of providing ecosystem benefits for humans. Figure 2 describes the interconnection of marine and coastal habitats the physical environment of which can be categorized into five different regimes such as the estuarine, the freshwater influenced, the near shore, the neritic and the oceanic. The changes and the alterations of different functions and processes of ecosystems result in changes to human welfare providing them with the beneficial final marine and coastal services (Luisetti *et al.*, 2010). It is necessary that these changes are being taken into account by the marine and coastal policy stakeholders. Some of the benefits or the final services derived by

intermediate services are the raw material and food, the biodiversity conservation, the greenhouse gas conservation, the coastal protection, the erosion control, the provision of amenity and recreation and the cultural heritage.

Numerous contingent valuation surveys have been conducted worldwide, including Greece, which try to estimate the final benefits derived from the integrated processes and functions of marine and coastal ecosystems such as water quality regulation. Water quality and aquatic life are affected directly by pressures provoked by mankind development in coastal and marine areas and indirectly by the lack of water quality management and implementation of policies.

Figure 2: The interrelation of marine and coastal habitats with the intermediate and final services.



Bockstael *et al.* (1989) carried out a research in Chesapeake Bay, the largest estuarine in USA which lies off the Atlantic Ocean, to estimate the participants' WTP concerning the changes in water quality improvements. In the survey, a hypothetical improvement in water quality from a current condition is considered and the data sets generated from three recreational activities such as beach use, boating and fishing have been analyzed using a Tobit model. Some years later in the same estuarine, Lipton (2003) utilized the contingent valuation approach, asking boaters to rank their perception of water quality. Factors seemed to influence the WTP, was the condition whether or not the boat was trailed or kept in the water and whether the vessel was a sail or powerboat. A Tobit model was applied to examine the WTP values; however, no demographic characteristics were included in the survey.

Hayes *et al.* (1992) examined the benefits of water quality improvement for residents in the Upper Narragansett Bay of Rhode Island in eastern USA. The variables under valuation are associated with the recreational activities, the general attitude towards water quality and the scenarios of water quality changes such as improvements that allow swimming and accessibility to the shell-fishing areas. Moreover, the analysis included questions about the distance from the estuary, the length of their residency, the year they settled in the area and general socioeconomic and demographic characteristics. The collected data as regards to the willingness to contribute to the improvement of the water quality of the estuary was examined using Logit model formulations.

As regards to contingent valuation surveys in Europe, Georgiou *et al.* (1998) by implementing regression models to analyse the derived data from the survey, estimated the value of eliminating the potential risks of illness caused by the quality of bathing water in two east coastal resorts such as Great Yarmouth and Lowestoft of the UK. Furthermore, they investigated the importance of inclusion of different factors such as the individual's acceptability to risk and health attitudes, except from the inclusion of traditional factors such as income, education etc. In the same territory, Georgiou *et al.* (2000) in order to enhance the previous results examined the WTP for the improvement of the quality of bathing water towards the risks of illness, to

provide policy makers with information about the preferences in terms of traditional socioeconomic variables, attitudes towards risk, trust and accountability of institutions and regulatory processes.

On the South-Western Scottish coastline, Hanley & Kriström (2002) conducted a contingent valuation approach utilizing the Kaplan-Meier survival curve to obtain a non-parametric estimate of the distribution function, concerning the replies of the participants to valuation questions. The WTP estimation has been explored by a modified Turnbull algorithm and for the additional examination of the data; a Tobit model was also applied. As the major bathing beaches along south-western Scottish coastline failed to meet the guideline standards of the Bathing Waters Directive,

Hanley *et al.* (2003) conducted another survey combining revealed preference with stated preference data to value coastal water quality improvements. Specifically, the utilization of revealed preference and contingent behavior models simultaneously, result in eliminating the hypothetical bias which is a common problem related to contingent valuation approach. There are also other studies referring to water quality improvement that utilize contingent behaviour data (Nahman & Rigby, 2008; Barry *et al.*, 2011).

In the estuarine system of Randers Fjord in Denmark, Atkins *et al.* (2007) carried out a study implementing the contingent valuation approach to examine public preferences for water quality improvements and specifically the reduced eutrofication. First the decision tree analysis has been applied to investigate the relationship between the respondent's decision and a number of variables such as the annual income, the distance of their residence from the Fjord. Next, willingness to pay bids was investigated via the decision tree analysis in order to reveal the complexity of respondents' preferences.

Goffe *et al.* (1995) tried to evaluate individuals' preferences towards water quality in the French Harbor of Brest. The corresponding WTP estimates for water health and preservation of the ecosystem against eutrofication has been examined by the application of Tobit model formulations.

In Guadiana Estuary between Portugal and Spain, Guimarães *et al.* (2011) report the first application of the contingent valuation approach to evaluate the public preferences as regards to the improvement of water quality. The survey employed socioeconomic variables, variables concerning the performance of leisure such as diving, sailing etc., or the performance of professional activity such as restaurants, hotels etc., the perception of water quality and environmental awareness variables. As far as the econometric methods are concerned, the voting behaviour was analyzed by the Logit model and bid decision by using Generalized Additive Models (GAM).

In the same estuary, Ramazo-Hernandez & Saz-Salazar (2012) investigating the respondent's preferences concerning the water quality improvements according to Water Framework Directive (WFD), utilized parametric and non-parametric approaches. Initially, a Logit model has been applied which was followed by a Spike model to deal with zero bids and secondly the Kriström's non parametric approach employed in order to find a robust estimate of mean WTP equal to \in 33 per family yearly. This estimate aggregated by the number of households residing in the Guadiana river basin (GRB) yielding a social benefit of improving water quality equal to \in 39 million per year. About half of the respondents (258 of the 505 respondents) stated that they were unwilling to pay extra money in their water bills to attain the water quality targets set by the WFD.

In Greece, although the literature on the valuation of indirect values is limited, there have been some surveys aiming at the determination of the role of

environmental valuation methodologies and derived information for policy makers. Kountouri *et al.* (2009) explore the value of building a wind farm using the contingent valuation. Moreover, Birol *et al.* (2006) studied the value of efficient water resources management policies with economic valuation such as cost-benefit analysis, the hedonic pricing method, the travel cost method, the choice experiment and the contingent valuation method. For the valuation of water quality, Birol *et al.* (2007) investigated the WTP of farmers for the adoption of an effective management of wastewater in the aquifer of Akrotiri in Cyprus.

As far as it concerns the surveys of the water quality of Greek marine and coastal areas, the literature is limited too (Kontogianni *et al.*, 2003; Jones *et al.*, 2008; Organtzi *et al.*, 2009; Halkos and Matsiori, 2011). Kontogianni *et al.* (2003) attempted to examine the impact of the quality of Thermaikos Gulf in Thessaloniki on the preferences of the residents. Respondents were asked to state their maximum willingness to contribute to the Gulf's restoration. The incorporated variables in the survey are related to the knowledge about the location of the discharged municipal wastes, the recreational activities such as fishing, swimming, walking in the coastline, sailing, the intended behaviour of respondents if the water quality improves and the motives contributing for such an improvement. Other variables included are related to economic valuation questions and socioeconomic characteristics. A logistic regression has been utilized in order to examine the payment principle, the attitude towards the intervention, the WTP estimates and selected motivations for improving the quality of the Gulf.

In the Northeast side of the Aegean Sea, where Mytilene is located, Jones *et al.* (2008) applied a contingent valuation survey in order to elicit respondents' WTP for the improvement of the coastal water quality resulting from the construction of a

Sewage Treatment Plant. Respondents were provided with questions as regards the most important problems in the area under consideration, the pollution factors, the attitude towards the construction of the plant and their preferences. Economic valuation section included the reasons to contribute to the improvement or not. Different econometric models have been applied in order to examine and compare the results of the survey, namely a multiple regression model, an exponential and a multiplicative model. Finally, a Tobit model has been utilized with the purpose of dealing with the zero WTP values.

In order to evaluate the environmental benefits derived from the construction of a wastewater treatment plant, Organtzi *et al.* (2009) administered a survey of the coast of Toroneos Gulf at the eastern side of the Cassandra peninsula in Greece. The questionnaire of the double bounded dichotomous choice contingent valuation approach included three groups of participants such as the permanent residents, the owners of a private holiday house and the campers who visit the seaside village of Kriopigi frequently. The variable inter alia of the demographic and socioeconomic characteristics were estimated using maximum likelihood.

Halkos & Matsiori (2012a, b) applied a contingent valuation survey in order to estimate the economic benefits derived from improved coastal quality of beaches along the Pagasitikos Gulf in central Greece. The objective of the survey is to investigate how the determinant factors of the coastal protection, the coastal development and coastal management affect the participants' in the survey WTP for the coastal zone quality improvement and the awarded beaches with a blue flag. The effect of the variables included, were analyzed via the implementation of OLS and Logit formulation resulting in the conclusion that a great number of respondents were willing to pay for improvements in coastal zones quality.

Author (year)	Goods/Services estimated	Country	Sample size/	Econometric Approach	WTP/WTA
			Elicitation Format		
Bockstael <i>et</i> <i>a1</i> . (1989)	Water quality improvement	Chesapeake Bay in USA	496 in person interviews	Tobit model	Average WTP of \$ (1000) 67,582
Goffe <i>et al.</i> (1995)	(1) Improved water health,(2) Preservation of the ecosystem against eutrophication	French Harbor of Brest	607 direct interviews	Tobit model	Mean WTP of FF 218 for health and FF 173 for ecosystem
Georgiou <i>et</i> <i>al.</i> (1998)	Bathing water quality	UK	400 in person interviews	Box-Cox procedure with semi- long form	Mean WTP of £12,64 per household
Georgiou <i>et</i> <i>al</i> . (2000)	Bathing water quality	UK	626 in person interviews	Multivariate statistical analysis	Mean WTP of £35.73 per household
Lipton (2003)	Water quality	Chesapeake Bay USA	Mail survey of 2510 sample units	Tobit model	Mean WTP for poor water quality rating: \$103; Fair water quality rating: \$ 124; Good: \$ 70; Very good: \$51;Excellent: \$ 38
Atkins <i>et al.</i> (2007)	Water quality improvements (reduced eutrofication)	Randers Fjord in Denmark	Mail survey of 207 respondents	Decision tree and regression analysis	Mean willingness- to-pay (in terms of a local tax) of DKK 57 (\in 7,64)per person, per month over a ten year period.
Guimarães et al. (2011)	Improvement of water quality	Guadiana Estuary between Portugal and Spain	67 face to face interviews in the pilot survey and 300 face to face interviews in the final survey	Two step model: (1) voting behaviour examined by logit (2) bid decision examined by Generalized Additive Models (GAM)	Mean WTP of € 47 per person per year over a five year period.

Table 1 [.]	CVM	Studies i	n marine	and coastal	l ecosystems'	water quality
Table 1.		Studies 1	II marme	and coasta	cosystems	water quanty.

Ramaio-	Water quality	Guadiana	505	Logit, Spike	Mean WTP of		
Hernandez	improvements	Estuary	personal	models and	€ 33 per		
and Saz-		between	interviews	Kriström's	family per		
Salazar		Portugal and		non-	vear		
(2012)		Spain		parametric	J Curr		
(2012)		Spuill		approach			
Contingent V	Valuation Studies	in Greece: Wa	ter quality in	marine and coas	tal ecosystems		
Contingent valuation Studies in Greece. Water quality in marine and coastal ecosystems							
Kontogianni	Impact of water	Thermaikos	480 face to	Logit to examine	Mean WTP of		
<i>et al.</i> (2003)	quality	Gulf in	face	the payment	5189		
		Thessaloniki	interviews	principle, the	drachmas per		
				WTP estimates	four monthly		
				attitude towards	water rates		
				intervention,	bıll		
				and selected			
				motivations for			
				improving			
T (1		т. 41	140	quality			
Jones et al.	Coastal water	In the	140	Regression	Mean WIP€		
(2008)	quality and the	northeast aide of the	interviewe	WTD	10.84 per		
	from operation	Aggeon See	linterviews	wir estimated and	avery 4		
	of the Sewage	in Mytilene		Tobit model	months over a		
	Treatment Plant	III Iviythene		to deal with	noticities over a period of 4		
	(STP)			protest	vears		
	(511)			responses	years		
Organtzi <i>et</i>	Environmental	Coast of	246	Maximum	Mean WTP of		
al. (2009)	benefit for the	Toroneos	personal	likelihood from	€ 56.40 per		
	construction of	Gulf at the	interviews	the double	respondent		
	wastewater	eastern side		bounded	Î		
	treatment	of the		dichotomous			
		Cassandra		choice model			
		Peninsula		(Hanemann, et			
				al., 1991).			
				Confidence			
				intervals are			
				constructed via			
				coefficients			
				trom the			
				Welbull			
Halkos and	Improvement	Pagasitikas	300 face to	OIS and	6 330/ and		
Matsiori	mprovement	Gulfin	face	Logit model	0.33% ware		
(2012a)	quanty (environment	Volos city	interviewe	formulation	2.3570 welt		
(2012a)	water	voios city	muer views	Tormulation	at the lowest		
	as well as				nrice of 5€		
	recreation				and at the		
	activities				highest price		
	uoti (11105)				hid of 50€		
					respectively.		

4. The economic foundations of contingent valuation

4.1 Welfare measures

Economic theory represents the basis for the formulation of the explicit goals of the non-market valuation. The standard neoclassical price theory is distinct from the non-market valuation, but it contributes to the development of the theoretical framework of this non-market assessment. The fundamental principle of neoclassical economic theory is the individuals' preferences' concerning goods and the element of preference ordering.

Preference ordering can be represented by a utility function, which is defined over goods, in our case the non-marketed ones. Market commodities can be expressed via the vector $X[x_1, x_2, x_3, ..., x_n]$ which is defined by the individuals and nonmarketed commodities via the vector $Q[q_1, q_2, q_3, ..., q_k]$ which is defined exogenously. The individuals' utility function U(X,Q) corresponds to a single number (X,Q). The utility function represents the individuals' preferences for a commodity A versus B if $U(X_A, Q_A) > U(X_B, Q_B)$. Another crucial element for the analysis is the money available to be spent on acquiring the desired items in the formal market depending on the system of the prices $P(p_1, p_2, ..., p_n)$.

The basic choice problem is the maximization of individuals' utility under the assumption of the available income I and the fixed levels of the non market commodities.

$$\max_{X} U(X,Q)$$

subject to $P * X \le I \quad Q \equiv Q_0$ (1)

Each market commodity has an optimal demand that can be expressed as $X' \equiv X(P,Q,I)$. Substituting the Marshallian demand¹ in the utility function we obtain the indirect utility which is a function of prices and income.

$$V(P,Q,I) \equiv \max_{X} U(X,Q)$$

subject to $P^*X \le I$ (2)

However, the individual's optimization problem can be expressed in the form of a utility maximization problem for a given constraint or as an expenditure minimization problem.

The minimum expenditure function m(P,Q,U) can be characterized by the duality to the indirect utility function and can be expressed by the problem:

$$m(P,Q,U) \equiv \min_{X} (P * X)$$

subject to $U(X,Q) \ge U$ (3)

When the direct and the indirect utility function are increasing and quasi-concave in Q, the minimum expenditure function is decreasing and convex in Q.

The aforementioned functions of indirect utility and minimum expenditure serve as a theoretical framework for the welfare estimation. Stated preference methods and the contingent valuation method specifically, can be considered as the estimation of the changes of the indirect utility and the expenditure function from the Q_0 situation to Q_1 . Interpreting the changes as individuals' utility we have two states such as $V_0 \equiv V(P,Q_0,I)$ and $V_1 \equiv V(P,Q_1,I)$. In order to estimate the change in terms of monetary values we use the two Hicksian measures of the compensating variation $CV \equiv CV(Q_0,Q_1,P,I)$ and the equivalent variation $EV \equiv EV(Q_0,Q_1,P,I)$. Compensating variation expresses the amount an individual has to deduct in order to

¹ The negative of the ratio of the derivatives of the indirect utility function produces the Marshalian curve (Haab *et al.*, 2002 pp. 6)

leave a person just as well off as in the initial situation after the change in income and prices. Adversely, equivalent variation expresses the amount of income paid in order to leave a person just as well off as the terminal situation (Haab *et al.*, 2002).

Another way to describe welfare measures is via the WTP and WTA measures. WTP is the maximum amount an individual contributes for an improvement of natural service or good as for example the water quality in marine ecosystems.

The relationship between *CV*, *EV*, *WTP* and *WTA* are described by the following equations:

$$WTP(Q_0, Q_1, P, I) \equiv CV(Q_0, Q_1, P, I) \quad if \ utility \ increase$$
(4)

$$WTP(Q_0, Q_1, P, I) \equiv -EV(Q_0, Q_1, P, I) \quad if \ utility \ decrease$$
(5)

$$WTA(Q_0, Q_1, P, I) \equiv EV(Q_0, Q_1, P, I) \quad if \ utility \ increase$$
(6)

$$WTA(Q_0, Q_1, P, I) \equiv -CV(Q_0, Q_1, P, I) \quad if \ utility \ decrease$$
(7)

The compensating and equivalent variation is also expressed by the expenditure function:

$$CV = m(P, Q_0, U_0) - m(P, Q_1, U_0)$$
(8)

$$EV = m(P,Q_0,U_1) - m(P,Q_1,U_1)$$
(9)

We can illustrate these concepts using the Box-Cox indirect utility function:

$$V_{Q} = a_{Q} + b_{Q}[(I^{\lambda} - 1)/\lambda] \quad if \quad Q = 0, 1 \quad a_{1} \ge a_{0} \quad b_{1} \ge b_{0}$$
(10)

This indirect utility function can be considered as a structure of a Constant Elasticity of Substitution utility function.

The compensating variation in terms of a Box-Cox indirect utility function can be written as:

$$CV = \{ [(b_0 I^{\lambda})/b_1] - [(\lambda a_1 - \lambda a_0)/b_1] - [(b_1 - b_0)/b_1] \}$$
(11)

These concepts imply the following relations:

$$\lambda > 1 \Rightarrow$$
 income elasticity of WTP < 0
 $\lambda = 1 \Rightarrow$ income elasticity of WTP = 0 (11a)
 $\lambda < 1 \Rightarrow$ income elasticity of WTP > 0

Willingness to pay (WTP) and willingness to accept (WTA) estimates considerably differ due to the income elasticity of the price of the environmental good or service under valuation implying that income elasticity depends upon the variation of Q (Carson & Hanemann, 2005; Flores and Carson, 1997).

4.2 Statistical and econometric issues in estimating WTP and WTA

The necessity of the WTP and WTA derives from the non-observable units of utility alterations since the welfare measures are monotone transformations of the utility function. Under the framework of the deterministic model of WTP/WTA for the environmental alteration we have to incorporate random units in order to combine the economic with the stochastic model. The first step is to estimate the WTP/WTA distribution and the second is to trace the probability distribution given the assumption of the utility maximization of the participants in the survey of the contingent valuation (Carson and Hanemann, 2005). For convenience we will use only the WTP measure to illustrate the modeling structure.

With the aim to obtain the distribution, WTP can be expressed by a linear regression. The expected value of the compensating surplus $E(CV) = \mu_{WTP}$ equates with the mean of WTP and a random term *e* which is normally distributed. That is:

$$WTP = \mu_{\rm WTP} + e \tag{12}$$

Alternatively the above equation can be expressed by the logarithm of WTP as:

$$\ln WTP = \mu_{\rm WTP} + e \tag{13}$$

Analyzing the WTP further, the individual' WTP function can be explained by the following form:

$$WTP_i = a_0 + a_1 Z_i + e_i \tag{14}$$

Where α_0 and α_1 represent the constant and the variance coefficient respectively; Z_i represents the vector of variance which includes individuals' socioeconomic variables; and e_i is a random term (with zero mean and variance σ^2) referring both to individuals' WTP random term and the additional error caused by the inefficient specification of the model from the researcher (Wang *et al.*, 2004).

The cumulative distribution function can be normal, logistic or Weibull and can be denoted as:

$$F(x) \equiv \Pr(X \le x) \equiv \sum_{k \le x} \Pr(X = k)$$
(15)

This expression represents the probability that a discrete random variable X can take values equal or smaller than a definite value of x. If we assume that the random variable is the compensating surplus the above equation can be transformed as:

$$F(x) \equiv \Pr(CV \le x) \equiv \sum_{k \le x} \Pr(CV = k)$$
(16)

The structure of the cumulative distribution function (cdf) is related to the type of the survey question. The respondent states the preferred value autonomously when is asked a question in an open-ended format. The probability of attaining the response as to the value, take the following form:

$$\Pr(CV = B) \equiv f(B) \tag{17}$$

Where f(B), represents the probability density function.

On the report of Hanemann (1984), in case of the closed-ended single-bound format the respondent is asked to choose to pay a value B or not for the change of the

status quo situation Q_0 to an alternative Q_1 . If the answer is 'yes', this implies the relation $CV \ge B$. The probability for attaining a positive answer is symbolized as:

$$\Pr(CV \ge B) \equiv 1 - F(B) \tag{18}$$

In the event of a negative response, this means that the compensating variation is less than the proposed value. That is:

$$\Pr(CV \le B) \equiv F(B) \tag{19}$$

Moreover, the closed-ended, double-bounded format includes two valuation questions regarding two proposed bids, B_{Higher} , B_{Lower} with the basic assumption that the second preferred value depends on the first one. Specifically, if the answer in the first question is positive then the second value is greater than the first. Adversely, if the first answer is negative then the second bid is smaller than the initial (Hanemann *et al.*, 1991). The different response combinations 'yes'/'yes', 'no'/'no', 'yes'/'no', 'no'/'yes', can be expressed in the form of response probabilities as:

$$\Pr(yes / yes) = \Pr(CV \ge B_{Higher}) = 1 - F(B_{Higher})$$
(20)

$$\Pr(yes / no) = \Pr(B_{Higher} \ge CV \ge B) = F(B_{Higher}) - F(B)$$
(21)

$$\Pr(no / yes) = \Pr(B \ge CV \ge B_{Lower}) = F(B) - F(B_{Lower})$$
(22)

$$\Pr(no/no) = \Pr(B_{Lower} \ge CV) = F(B_{Lowet})$$
(23)

Another way to obtain the distribution of WTP, is by applying the random utility maximization approach (RUM) (Hanemann, 1984; Mitchell *et al.*, 1989; McFadden, 1974; Ben-Akiva *et al.*, 1985). In the framework of (RUM), the direct utility $U_{ij}(.)$ is equal to the indirect $V_{ij}(.)$ and the random term e_{ij} .

$$U_{i}(I_{j}, Z_{j}, Q; e_{ij}) = V_{i}(I_{j}, Q, Z_{j}) + e_{ij}$$
(24)

Where i = 0,1 describing the different states as 0 for the initial situation or status quo and 1 for the final state; I_j represents the income of the individual; Z_j the household characteristics of the individual *j*, Q_i the change of the environmental good or amenity and e_{ij} the random term. The basic assumption of RUM model is that the respondent can answer with certainty about the contribution for an environmental change, however, the researcher treats the individual's preferences as random (Carson & Hanemann, 2005; Haab *et al.*, 2002; Giraud *et al.*, 2001; Cooper *et al.*, 2002).

The corresponding response probability function of the RUM model can be expressed as:

$$\Pr(yes_{i}) = \Pr(V_{1} \ge V_{0}) = 1 - F(B)$$
(25)

Where $V_1 = (Q_1, Z_j, I_j - B, e_{1j})$ and $V_0 = (Q_0, Z_j, I_j, e_{0j})$ denote the indirect utilities of the item under estimation at the initial 0 and final 1 situation.

The aforesaid response probability function constitutes the starting point for the parametric and non-parametric analysis. Initially, we can illustrate some of the different parametric approaches depending on the basic form of random utility function. It is well acknowledged that the utility in RUM specifications is characterized by the sum of a deterministic and a random term which is expressed by the equation $e_{ij} = e_{1j} - e_{0j}$. Further, the random terms are independently and identically distributed (IID) with zero mean.

When we take for granted the linearity with respect to income, we assume constant marginal utility across the individuals. In this case, the random utility function takes the form:

$$V_{ij}(I_j) = a_i Z_j + b_i I_j$$
⁽²⁶⁾

The above equation can be transformed into the following forms of the initial state 0 and the final 1:

$$V_{0j}(I_j) = a_0 Z_j + b_0 I_j$$
(27)

$$V_{1i}(I_i - B) = a_1 Z_i + b_1 (I_i - B)$$
(28)

The term *B* represents the bid amount offered to the respondents for the implemented measure corresponding to the environmental change.

The difference between the indirect utilities given that $a = a_1 - a_0$ and $b_1 = b_0$ is given as:

$$V_{1j}(I_j - B) - V_{0j}(I_j) = (a_1 - a_0)Z_j + b_1(I_j - B) - b_0I_j = = aZ_j + bB$$
(29)

Having specified the deterministic component of the random utility model with linear income we can also identify the structure of the response probability function, as:

$$Pr(yes) = Pr(aZ_{j} - bB + e_{ij} > 0) = Pr(e_{ij} < aZ_{j} - bB)$$
(30)

The corresponding standard normal probability can be designed as:

$$\Pr(yes) = \Pr(e_{ij} < aZ_j - bB) = \phi[(aZ_j / \sigma) - (bB / \sigma)]$$
(31)

Where ϕ , denotes the cumulative distribution response probability function of the standardized errors.

If the random term is characterized by the symmetric logistic distribution

 $e \sim \left(0, \frac{\pi^2 \sigma}{3}\right)$, we can design the Logit model as:

$$\Pr(yes) = \Pr[(1 + \exp(-(aZ_j / \sigma - bB / \sigma)))]^{-1}$$
(32)

If the random component is standard logistic distributed $e \sim \left(0, \frac{\pi^2}{3}\right)$, the Probit model can be estimated as:

$$\Pr(yes) = \phi[(aZ_j / \sigma) - (bB / \sigma)]$$
(33)

Another parametric approach of the RUM is the log-linear form as regards the income. The basic assumption is that the marginal utility $\frac{\partial V_{ij}}{\partial I_j} = \frac{b}{I_j}$ is not constant across the individuals but adversely is decreasing given b > 0. The response

probability expression with normal distributed errors can be recognized as:

$$\Pr(yes) = \Pr\left[\left(\frac{aZ_j}{\sigma} + \frac{b\ln[(\frac{I_j - B}{I_j})]}{\sigma}\right) > -e_{ij}\right]$$
(34)

The corresponding standard normal cumulative response probability form can be calculated as:

$$\Pr(yes) = \phi \left(\frac{aZ_j}{\sigma} + \frac{b\ln[(\frac{I_j - B}{I_j})]}{\sigma} \right)$$
(35)

The Box-Cox version of random utility model can be represented by the following equation:

$$V_{ij} = a_i Z_j + b I_{ij}^{\ \lambda} + e_{ij} \tag{36}$$

where $I^{\lambda} = \frac{I^{\lambda} - 1}{\lambda}$. The random utility function can take different forms as far as the term λ takes different values:

$$\lambda = 1 \Longrightarrow I^{\lambda} = I - 1$$

$$\lambda = 0 \Longrightarrow I^{\lambda} = \ln I$$

$$\lambda = -1 \Longrightarrow I^{\lambda} = bI^{\lambda - 1}$$

$$\lambda = 2 \Longrightarrow I^{\lambda} = \frac{I^{2} - 1}{2}$$
(37)

The response probability functions when the transformation term is constant $\lambda = 1 \Longrightarrow I^{\lambda} = I - 1$ is given by:

$$\Pr(yes) = \Pr\left[aZ_{j} + b\left[\frac{(I_{j} - B)^{\lambda} - I_{j}^{\lambda}}{\lambda}\right] > -e_{ij}\right]$$
(38)

The corresponding cumulative form can be expressed as:

$$\Pr(yes) = \phi \left[\frac{aZ_j + b \left[\frac{(I_j - B)^{\lambda} - I_j^{\lambda}}{\lambda} \right]}{\sigma} \right]$$
(39)

The following equation represents the response probability distribution function that belongs to the family of parametric approaches with the difference that utilizes an asymmetric distribution to some extent such as the Weibull distribution.

$$Pr(yes) = \exp(-\exp(aZ_i + bB))$$
(40)

The confidence intervals are produced via the coefficients from the Weibull function (Krinsky & Robb, 1986; Park *et al.*, 1991).

The proper formulation of the response probability function aids to the estimation of the welfare measures expressed by the central tendencies such as mean and median of the WTP. The employment of the central tendencies depends on the specific criteria applied by the decision maker (Bateman *et al.*, 2002).

The majority of the researchers in the field of modelling the non-market values through the contingent valuation approach employ parametric models, such as the Probit and Logit, especially to investigate data derived from the discrete choice models. The distribution of the Logit and Probit model serves as a means of estimating the mean WTP. Nevertheless, the response probability function can either be parametric or non-parametric. The parametric approaches use the response probability function as a known function, while the distribution of the WTP depends on a vector that constitutes a finite number of unknown parameters (An, 1996). Modelling the WTP distribution via parametric approaches results in efficient estimators, in appropriate behaviour description, or in the derivation of precise estimates of central tendencies, although, there are serious disadvantages such as the biased and inconsistent estimates due to the strong interconnection to distributional and functional assumptions (Creel & Loomis, 1997; Osorio & Mittelhammer, 2012; Crooker & Herriges, 2004).

The underlying problems of parametric methods have led to the nonparametric approaches. One of the main purposes of the non parametric estimators is the investigation of the distribution of the response probability or survivor function. The distribution's shape of the survivor function is tracted out by a discrete set of points (Carson & Hanemann, 2005). There have been introduced pioneering methods, in order to tract out the discrete set of points, that the mean WTP is independent to a given distribution of the model implemented such as Logit or Probit (Creel & Loomis, 1997; Kriström, 1990; Turnbull, 1976; Giraud *et al.*, 2001).

One of the ways to connect the discrete set of points that depicts the shape of the survivor function is the Turnbull distribution-free estimator. The estimator, accomplish to provide a lower bound of the mean WTP given that $\phi_n \leq \phi_{n+1} if B_n \leq B_{n+1}$ where ϕ denotes the cumulative density function (cdf) and *n* the number of bid values (*B*) offered (Crooker & Herriges, 2004). In this way we have:

$$E(WTP)_{LB} = \sum_{n=1}^{N} B_n(\phi_{n+1} - \phi_n) \le E(WTP)$$
(41)

The application of the linear interpolation, is another easy trial to develop the exact shape of the distribution of the response probability function according to Kriström (1990), which results in a piecewise linear cdf under the assumption that the probability of the Bids offered are uniformly distributed. The upper bound of the WTP expected value given $\phi_0 = 1$ and $\phi_{n+1} = 0$ is expressed as:

$$E(WTP)_{UB} = \sum_{n=1}^{N} (B_n + B_{n+1})(\phi_n - \phi_{n+1})$$
(42)

Smoothing assumptions concerning the distribution leads to a more consistent description of the shape of the survivor function distribution. What is more, semi-nonparametric estimators allow the inclusion of descriptive variables apart from the bid price (An, 1996; Araña & Leon, 2005).

After the Chen & Randall (1997) estimator, the compensating variance function is separated into a nonstochastic WTP function such as the exponential form and a random term, so that *CV=non-stochastic(WTP)-e*. Using the Gallant Fourier Flexible Form to model the nonstochastic component, implies a monotonic transformation of the error term. One of the advantages of Chen and Randall's estimator is that it allows the implementation of the standard maximum likelihood procedure (Creel & Loomis, 1997; Crooker & Herriges, 2004).

Creel & Loomis (1997) estimator is almost similar to the aforementioned although, it differs in terms of the utilization of the utility difference as in equation (29) and the necessity of the estimation of only one series.

There are also plenty of surveys that introduce semi- nonparametric approaches including smoothness assumption (An, 2000; Horowitz, 1992; Cooper *et al.*, 2002; Khan *et al.*, 2012; Belluzzo, 2004; Huang *et al.*, 2008; Rothe, 2009; Fezzi & Ferini, 2012), but have not so far being applied thoroughly in empirical research either because of the required knowledge in the field or the difficulty in reference with the definition of the particular smoothing assumptions. In the recent literature, as to valuation of marine and coastal goods and services, Landry & Liu (2009) utilize a

semi parametric approach in order to estimate revealed and stated preference recreation demand models concerning the beach in North Carolina.

This approach is an alternative versus the parametric approach in terms of misspecification relating to the unobserved heterogeneity. Madani *et al.* (2012) applied a semi non-parametric distribution free estimator to analyze the data collected from the contingent valuation approach respecting the value of coral reef within Kish Island. Yet, non-parametric estimators in marine and coastal empirical research are extensively used compared to semi non-parametric estimators (Edwards, 2009; Van Biervliet & Nunes, 2006; Petrolia & Kim, 2009; Casiwan-Launio *et al.*, 2011; Jørgensen *et al.*, 2012).

5. A critical consideration of contingent valuation method

Although contingent valuation is dominant among environmental methodologies, there are controversies related to the validity and reliability of the estimates. The validity of the results refers to the consistency and the reproducibility of results (Kealy *et al.*, 1990). The validity can be divided into content, structure and criteria validity. *Content validity* refers to the quality of the tools used, such as the questions involved in the survey. As noted by Bateman *et al.* (2002) the important issues as regards to the content validity is whether the appropriate questions is utilized in order to elicit the stated preferences of the participants or the participant has answered in the offered question. Evidently, if it is asked the wrong question, this leads to invalid estimation.

The *structure validity* refers to the ability of the theoretical structure of the hypothetical market to estimate the real economic value which is stated by the respondents (Freeman, 1993). The structure validity can be further divided into two

categories. The first category relates to the *convergent validity*, in which the estimates are compared with others to assess the degree of convergence as envisaged by the theory. The second category refers to the *theoretical validity* in which the results are examined in the context of their consistency with the rules of economic theory. The theoretical validity involves the estimation of the WTP and the examination of the relationship between the estimates and standard economic variables such as income (Mitchell & Carson, 1989).

The *criterion validity* is relevant to the comparison of a measure with another which is considered that is the most consistent within the theoretical structure. For example, the validity of the questions to assess the Hicksian surplus can be estimated by comparing the results with a hypothetical structure of questions considered to be the most associated to the theoretical structure of the method used (Bateman *et al.*, 2002).

Another considerable issue is the reliability which relates to the structure and the internal consistency of the contingent valuation approach. Hanley (1997) proposed two ways to estimate the reliability of the survey results. Initially, it is necessary to test the convergent validity and secondly to carry out the test-retest method applied by Loomis (1990), where the same questions were repeated to the same respondents in order to assure the consistency of their responds. On the report of Whitehead *et al.*, (1995) another application constitutes of two similar questions in different formats so as to check the positive correlations between the measures.

Apart from the reliability the elicitation effect is a controversial issue too. The value of an environmental amenity is elicited via different elicitation formats such as open ended questions, bidding games, payment cards, single –bounded dichotomous choice questions and multiple-bound dichotomous choice questions (Chien *et al.*,

2005), whereas the most common formats include 'response effects'. Starting point bias is a difficulty in terms of a problematic initial bid that has to encounter the utilization of the bidding game format, the single bounded dichotomous choice format and the double-bound dichotomous choice format (Venkatachalam, 2004). The difficulty with the starting point bias refers probably to the lack of the researcher's experience or the respondents' familiarity to commodities under examination, or the respondents' impatience concerning the bidding process (Silberman & Klock, 1989).

Efforts to face the problem relate to particular design of the questionnaire. Using a payment card is a solution; however, this may lead to anchoring in which the design of a numerical bid can result in an alteration of the response. Besides, explicit models can be generated to reduce the negative effects. Aprahamian *et al.* (2007) proposed a model with an open-ended follow-up question and an anchoring random parameter. Chien *et al.* (2005) includes the 'yea says bias' in the model which refers to the biased responses of participants regardless the content of the question. The 'yea saying' problem is frequently related to 'fat right tail problem' where the cumulative density functions are disproportionately large leading to large mean estimates of WTP (Ready & Hu, 1995).

The information as regard with the good or service to be estimated is crucial to the estimates of the mean WTP/WTA. According to Bergsrtom *et al.* (1990) the type of information may affect seriously the results either in a positive or in a negative way. The required information includes the quality of the environmental amenity, the budget constraints, the relative expenditures, or the knowledge about other substitutes (Bergstrom *et al.*, 1989; Blomquist & Whitehead, 1998).

Embedding effect is another problem among a great list. As claimed by Kahneman & Knetsch (1992) there is a tendency to overrate a good if it is estimated

on its own rather than if it is part of a more inclusive agenda. The embedding effect has been interpreted in terms of part-whole bias, disaggregation bias and sub-additivity, scope and order effects. Hanemman (1994) redefined the concept of the embedding effect proposing the further classification in the sequencing or question order effect which is related to the order of questions, the scope effect which is related to the valuation of equally different sizes of a presented good under estimation and the sub-additivity effect which is referred to as the 'adding up' problem (Dupont, 2003).

There is a debate concerning the existence of scope effect, however, the empirical research that considers this problem is very limited (Venckatachalam, 2004). As suggested by Mitchell & Carson (1989), the embedding effect can be minimized by the better interpretation of the presented commodity to respondents via all different types of media including optical.

Strategic behaviour is a common problem that characterizes the contingent valuation approach and especially the open-ended question format. The respondents can be classified as regards to their attitude to act strategically and not reveal their real intention to pay for the change of the environmental amenity. Specifically, free riders are those that underestimate the good under consideration given that there are always others willing to contribute for the change. The second category of the strategic behaviour refers to the overpledging in terms of overstating the WTP given that an external source will provide the financial support for the provision of the good or service under estimation (Mitchell & Carson, 1989).

Additionally, strategic behaviour can be revealed by the continuous same response during the survey which leads to flattened WTP distribution (Carson & Hanemann, 2005). Although, Samuelson (1954), has commented upon the design of the contingent valuation approach that communicate the problem of strategic behaviour, for some other researchers, strategic behavior does not constitute a crucial problem (Cummings *et al.*, 1986; Mitchell & Carson, 1989; Griffin *et al.*, 1995; Schulze *et al.*, 1981). Whittington *et al.* (1992) utilizing a multivariate analysis examined the impact of the existence of time to think and revise the WTP responses in order to investigate the existence strategic bias. Deshazo (2002) developed a head-to-head test to explore the strategic bias, the yea-saying and the anchoring problem concluding that it is more efficient using the iterative bid design in order to minimize the problems.

In addition, researches have to take into account the true zero bids and protest responses. In the majority of the contingent valuation studies there are participants that are not willing to contribute for an environmental improvement of a public good either because they cannot afford to or they consider unfair the payment vehicle, or they adopt a free riding behavior or they do not possess the sufficient information (Jørgensen *et al.*, 1999; Strazzera *et al.*, 2003). One of the solutions to the specific problem is the exclusion of the protest zero bids of the sample if the respondents state that they are sincerely indifferent to the good under estimation, whereas this can lead to another problem named 'sample selection bias' in terms of flatness of the likelihood function.

Alternative solution is the utilization of Tobit model (Tobin, 1958), the use of the sample selection models such as Heckman (Heckman, 1979), or the use of the double hurdle models (Cho *et al.*, 2008). Recently, Halkos & Jones (2012), with the purpose of determining the effect of social factors on the individuals' state concerning the contribution for an improvement of environmental protection of biodiversity, utilized the Heckman sample selection model and the double hurdle model in order to tackle the problem of the refusals to pay and the protesters to respond. In the same line, Halkos & Matsiori (2012a), examining the current and potential economic value of an artificial lake, applied the double hurdle model in comparison with Tobit model to deal with zero responses, while Tobit model fail to take into account the sources of the zero responses.

The hypothetical scenarios of payment and provision of the good or service that is the basic feature of contingent valuation survey, can lead to a divergence of respondent's statement and thus to a hypothetical bias. As stated by Ajzen (2004) the reason for the existence of the hypothetical bias is the different intentions of respondents in a hypothetical or in a real frame of reference. Empirical results in the literature show a higher WTP in a hypothetical context rather than in an actual (Cummings *et al.*, 1995; Brown *et al.*, 1996; Kealy *et al.*, 1990; List & Gallet, 2001; Harrison, 2006).

One of the approaches to deal with the hypothetical bias has been proposed by Cummings & Taylor (1999). 'Cheap talk approach' includes an integrated discussion with the respondent concerning the hypothetical bias problem. Another approach proposed is the 'certainty approach' which depends on the respondent's certainty to contribute to the change of the amenity (Murphy *et al.*, 2005; Champ & Bishop, 2001; Poe *et al.*, 2002; Vossler *et al.*, 2003; Blumenschein *et al.*, 2008).

6. A summary of the main points

Humans' well being, basic needs and economy rely upon the exploitation of the most benefits provided by the ecosystems. However, the economic activity results to damages to the biodiversity, the water quality and the sea level rise. Hence, to manage the adverse consequences it is important to develop the knowledge concerning the valuable goods and services that various ecosystems provide. Although contingent valuation is considered as one of the most promising methods there are various limitations that have to be addressed carefully, either by the inclusion of an appropriate question format, or by conducting the analysis of the data with a proper econometric method parametric, non-parametric or semi nonparametric.

The parametric approaches treat the response probability function as a known function, while the distribution of the WTP depends on a vector that constitutes a finite number of unknown parameters. Moreover, estimators such as the nonparametric Turnbull and Kriström's, have been applied in the literature. The semi non-parametric estimators which include smoothing assumptions concerning the distribution lead to a more consistent depiction of the shape of the survivor function distribution as they allow the inclusion of descriptive variables apart from the bid price. Although there are also plenty of surveys that introduce semi non-parametric approaches, there have not been applied thoroughly in empirical research or in the field of marine and coastal goods and services.

To criticize the approach of contingent valuation a number of issues were presented such as the validity and reliability of the estimates. The validity of the results refers to the consistency and reproducibility of results and the reliability relates to the structure and the internal consistency of the contingent valuation approach. Another consideration is the elicitation effect that is related to the estimation of the value of an environmental amenity via different elicitation formats. Starting point bias is an included disadvantage in terms of a problematic initial bid.

Furthermore, given that the contingent valuation is designed in harmony with hypothetical scenarios of payment and provision of the good or service, the respondents alter their statements. Another common problem is the limited available

information of the respondents and its crucial role to the estimates of the mean WTP/WTA. Concerning the voting behaviour, the respondents can be classified as free riders who underestimate the WTP, those who overstate the WTP given external financial support and as protesters that do not agree to contribute for an environmental improvement of the good or service under estimation.

To date the literature of the contingent valuation studies in the field of marine and coastal ecosystem is consistently increasing. The challenge is to succeed the elicitation of an appropriate number of response units in order to estimate the welfare measures dealing at the same time with the majority of included bias in the research. Besides of reviewing the available studies concerning the water quality of marine and coastal systems, the scope of the present paper is to provide a linkage between the important notions of value, the economic theory which is based the contingent valuation approach in accordance with the statistical models to derive the welfare measures and included disadvantages of the method.

References

Adamowicz, W., Boxall, P., Williams, M., and Louviere J., (1998). Stated Preferences Approaches for Measuring Passive Use Values: Choice Experiments and Contingent Valuation, *American Journal of Agricultural Economics*, 80: 64–75.

Ajzen, I., Brown, T.C. and Carvajal, F., (2004). Explaining the discrepancy between intentions and actions: the case of hypothetical bias in contingent valuation, *Personality and Social Psychology Bulletin*, 30: 1108–21.

Alias, R., Mohd Rusli, Y., Juwaidah, S., (2008). Use of Dichotomous Choice Contigent Valuation Method to Value the Putrajaya Weatland Park, *International Journal of Management Studies*, 15(2):81-97.

An, M.Y., (1996). Semi parametric Estimation of Willingness to Pay Distributions. Department of Economics. Working Paper 96-20. Duke University.

An, M. Y., (2000). A semiparametric distribution for willingness to pay and statistical inference with dichotomous choice contingent valuation data, *American Journal of Agricultural Economics*, 82: 487–500.

Aprahamian, F., Chanel, O., Luchini, S., (2007). Modelling starting point bias as unobserved heterogeneity in contingent valuation surveys: an application to air pollution, *Journal of Agricultural Economics*, 89: 533–547.

Araña, Jorge E., and León, Carmelo J., (2005). Flexible mixture distribution modeling of dichotomous choice contingent valuation with heterogeneity, *Journal of Environmental Economics & Management*, 50: 170–178.

Arrow, K. J. and Fisher, A., (1974). Environmental preservation, uncertainty and irreversibility, *Quarterly Journal of Economics*, 89: 312-319.

Atkins, J.P., Burdon, D. and Allen J.H., (2007). An application of contingent valuation and decision tree analysis to water quality improvements, *Marine Pollution Bulletin*, 55: 591-602.

Báez, A., Niklitschek, M., and Herrero, L. C., (2009). The valuation of historical sites: a case study of Valdivia, Chile, *Journal of Environmental Planning and Management*, 52: 97–109.

Barbier, E., Hacker, S., Kennedy, C., Koch, E., Stier, A., Silliman, B., (2010). The value of estuarine and coastal ecosystem services, *Ecological Monographs*, 81:1521–1531.

Barry, L., van Rensburgn, T.M., Hynes, S., (2011). Improving the recreational value of Ireland's coastal resources: A contingent behavioural application, *Marine Policy*, 35: 764–771.

Bateman, I.J., Carson, R.T., Day, B., Hanemann, W.M., Hanley, N.D., Hett, T., Jones-Lee, M.W., Loomes, G., Mourato, S., Özdemiroglu, E., Pearce, D.W., Sugden, R., Swanson, J., (2002). *Economic Valuation with Stated Preference Techniques: A Manual*. Edward Elgar Publishing, Cheltenham.

Bateman, I., Lovett, A., Brainard, J., (2003). *Applied Environmental Economics*. Cambridge University Press, Cambridge.

Belluzzo, W., (2004). Semiparametric approaches to welfare evaluations in binary response models, *Journal of Business and Economic Statistics*, 22: 322-330.

Ben-Akiva, Moshe, Lerman S.R., (1985). *Discrete Choice Analysis: Theory and Application to Travel Demand*, Cambridge, MA: MIT Press.

Bergsrtom, J.C., Stoll J.R., Randall, A., (1990). The impact of information on environmental commodity valuation decisions, *American Journal of Agricultural Economics*, 72:614–21.

Bergstrom, J.C., Stoll, J.R., Randall, A., (1989). Information effects in contingent markets, *American Journal of Agricultural Economics*, 71: 685–691.

Birol, E., Kountouri, P., Kountouris, Y., (2007). Farmers' Demand for Recycled Wastewater in Cyprus: A contingent valuation approach, Environmental Economy and Policy Research, Discussion Paper Series, No. 24, University of Cambridge.

Birol, E., Karousakis, K., Kountouri, P., (2006). Using economic valuation techniques to inform water resources management: A survey and critical appraisal of available techniques and an application, *Science of the Total Environment*, 365: 105-122.

Blomquist, G.C., Whitehead, J.C., (1998). Resource Quality Information and Validity of Willingness to Pay in Contingent Valuation, *Resource and Energy Economics*, 20: 179–196.

Blumenschein, K., Blomquist, G.C., Johannesson, M., Horn, N. & Freeman, P., (2008). Eliciting willingness to pay without bias: Evidence from a field experiment, *Economic Journal*, 118(1): 114-137.

Bockstael, N.E., K.E., McConnell and Strand, I.E., (1989). Measuring the benefits of improvements in water quality: The Chesapeake Bay, *Marine Resource Economics*, 6 (1): 1–18.

Bockstael, N.E., Freeman, A.M., Kopp, R.J., Portney, P.R., Smith, V.K., (2000). On measuring economic values for nature, *Environmental Science and Technology*, 34: 1384–1389.

Borghi, J., Jan, S., (2008). Measuring the Benefits of Health Promotion Programmes: Application of Contingent valuation Method, *Health Policy*, 87: 235-248.

Boxall, P., Adamowicz, W., Swait, J., Williams, M., Laviere, J., (1996). A comparison of stated preference methods for environmental valuation, *Ecological Economics*, 18: 243–253.

Brown, T.C., Champ, P.A., Bishop, R.C., McCollum D.W., (1996). Which response format reveals the truth about donations to a public good?, *Land Economics*, 72:152–66.

Burke, L., Kura, Y., Kassem, K., Revenga, C., Spalding, M., McAllister, D., (2001). *Pilot Analysis of Global Ecosystems: Coastal Ecosystems*. Washington, DC, USA: World Resources Institute p. 77.

Carson, R., (2000). Contingent Valuation: A User's Guide, *Environmental Science and Technology*, 34: 1413-1418.

Carson, R.T., Hanemann, M.W., (2005). *Contingent valuation*. In: Maler, K.-G., Vincent, J.R. (Eds.), Handbook of Environmental Economics, vol. 2. Elsevier/ North-Holland, New York (Chapter 17).

Casiwan-Launio, C., Shinbo, T., Morooka Y., (2011). Island villagers' willingness to work or pay for sustainability of a marine fishery reserve: the case of San Miguel Island, Philippines, *Coastal Management*, 39: 459–477.

Champ, P.A., Bishop R.C., (2001). Donation Payment Mechanisms and Contingent Valuation: An Empirical Study of Hypothetical Bias, *Environmental and Resource Economics*, 19: 383-402.

Chen, H., Randall A., (1997). Semi-nonparametric estimation of binary response models with an application to natural resource valuation, *Journal of Econometrics*, 76: 323-340.

Chien, Y.L., Huang, C.J., Shaw, D., (2005). A general model of starting point bias in double-bounded dichotomous contingent valuation surveys, *Journal of Environmental Economics and Management*, 50: 362–377.

Cho, S.-H., S.T., Yen, J.M., Bowker, and Newmark, D.H., (2008). Modeling Willingness to Pay for Land Conservation Easements: Treatment of Zero and Protest Bids and Application and Policy Implications, *Journal of Agricultural and Applied Economics*, 40:267–85.

Christie, M., Azevedo, C., (2002). *Testing the consistency in benefit estimates across contingent valuation and choice experiments: a multiple policy option application*, Paper presented at the 2nd World Congress of Environmental and Resource Economists, Monterrey, California, June.

Cicchetti, C.J., Wilde, L.L., (1992). Uniqueness, irreversibility, and the theory of nonuse values, *American Journal of Agricultural Economics*, 74: 1121–1125.

Ciriacy-Wantrup S.V., (1947). Capital return from soil conservation practices, *Journal of Farm Economics*, 29: 1188-90.

Cohen, J.E., Small, C., Mellinger, A., Gallup, J., and Sachs, J., (1997). Estimates of coastal populations, *Science*, 278: 1211–1212.

Cooper, J.C., Hanemann, M., and Signorello, G., (2002). One-and-One-Half-Bound Dichotomous Choice Contingent Valuation, *The Review of Economics and Statistics*, 84:742-750.

Costanza, R., (2000). Social goals and the valuation of ecosystem services, *Ecosystems*, 3: 4–10.

Costanza, R., (2006). Nature: ecosystems without commodifying them, *Nature*, 443:749-749.

Creel, M., Loomis, J., (1997). Semi-nonparametric Distribution-Free Dichot-omous Choice Contingent Valuation, *Journal of Environmental Economics and Management*, 32: 341-358.

Crooker, J.R., Herriges, J.A., (2004). Parametric and semi-nonparametric estimation of willingness-to-pay in the dichotomous choice contingent valuation framework, *Environmental and Resource Economics*, 27: 451–480.

Cummings, R.G., Brookshire, D.S., Schulze, W.D., (1986). *Valuing environmental goods: a state of the arts assessment of the contingent valuation method*. Totowa, NJ: Roweman and Allanheld.

Cummings, R.,G., Harrison, and Rustrom E., (1995). Homegrown Values and Hypothetical Surveys: Is the Dichotomous Choice Approach Incentive Compatible?, *American Economic Review*, 85: 260-266.

Cummings, Ronald G., Taylor, Laura O., (1999). Unbiased Value Estimates for Environmental Goods: A Cheap Talk Design for the Contingent Valuation Method, *American Economic Review*, 89: 649-65.

Daily, G.S., Alexander, P.R., Ehrlich, L., Goulder, J., Lubchenco, Matson, P.A., Mooney, H.A., Postel, S., Schneider, S.H., Tilman, D., Woodwell, G.M., (1997). Ecosystem Services: Benefits Supplied to Human Societies by Natural Ecosystems, *Issues in Ecology*, 2:1-18.

De Groot, R., Wilson, M., Boumans, R., (2002). A typology for the description, classification and valuation of Ecosystem Functions, Goods and Services (p. 393-408). In: "The Dynamics and Value of Ecosystem Services: Integrating Economic and Ecological Perspectives", *Ecological Economics*, 41: 367-567.

DeShazo, J.R., (2002). Designing transactions without framing effects in iterative question formats, *Journal of Environmental Economics and Management*, 43: 360–385.

Diener, A., O'Brien, B., Gafni, A., (1998). Health care contingent valuation studies: A review and classification of the literature, *Health Economics*, 7: 313-326.

Dupont, D.P., (2003). CVM embedding effects when there are active, potentially active and passive users of environmental goods, *Environmental & Resource Economics*, 25: 319–341.

Edwards, P., (2009). Sustainable financing for ocean and coastal management in Jamaica: The potential for revenues from tourist user fees, *Marine Policy*, 33: 376–385.

Eom, Y-S., Larson, D.M., (2006). Improving environmental valuation estimates through consistent use of revealed and stated-preference information, *Journal of Environmental Economics and Management*, 52: 501-516.

Farber, S., Costanza, R., Wilson, M., (2002). Economic and ecological concepts for valuing ecosystem services, *Ecological Economics*, 41: 375–392.

Fezzi C., and Ferrini S., (2012). *Non market valuation via Generalized Additive Models*, *CSERGE* School of Environmental Sciences, University of East Anglia, Norwich, NR4 7TJ, UK.

Fisher, B., Turner, R.K., Morling, P., (2009). Defining and classifying ecosystem services for decision making. *Ecological Economics*, 68(3): 643–653.

Fletcher, S., Saunders, J. & Herbert, R.J.H., (2011). A review of the ecosystem services provided by broad-scale marine habitats in England's MPA network, *Journal of Coastal Research*, 64: 378 – 383.

Flores, N.E., Carson, R.T., (1997). The Relationship Between the Income Elasticities of Demand and Willingness to Pay, *Journal of Environmental Economics and Management*, 33: 287–295.

Freeman A.M., III (1993). *The measurement of environmental and resource values*, Washington, DC: Resource for the Future.

Georgiou, S., Langford, I.H., Bateman I., Turner R.K., (1998). Determinants of Individual Willingness to Pay for Reductions in Environmental Health Risks: A Case Study of Bathing Water Quality, *Environment and Planning A*, 30: 577–594.

Georgiou, S., Langford, I.H., Bateman I., Turner R.K., (2000). Coastal Bathing Water Health Risks: Assessing the Adequacy of Proposals to Amend the 1976 EC Directive, *Risk, Decision and Policy*, 5: 49–68.

Giraud, K.L., Loomis, J.B., Cooper, J.C., (2001). A Comparison of Willingness to Pay Estimation Techniques from Referendum Questions, *Environmental and Resource Economics*, 20: 331–346.

Goffe, P., (1995). The benefits of improvements in coastal water quality: a contingent approach, *Journal of Environmental Management*, 45: 305–317.

Gommes, R., du Guerny, J., Nachtergaele, F., Brinkman, R., (1998). *Potential Impacts of Sea-Level Rise on Populations and Agriculture*. Food and Agriculture Organization of the United Nations, SD (Sustainable Development) Dimensions/Special.

Goulder, L.H., Donald, K., (1997). *Valuing ecosystem services: philosophical bases and empirical methods*. In: Daily, G.C. (Ed.), Nature's Services: Societal Dependence on Natural Ecosystems. Island Press, Washington, DC, p. 23–48.

Goulder, L., Kennedy, D., (2010). *Interpreting and estimating the value of ecosystem services*. In P. Kareiva, G. Daily, T. Ricketts, H. Tallis, and S. Polasky, editors. The theory and practice of ecosystem service valuation and conservation. Oxford University Press, Oxford, United Kingdom.

Griffin, C.C., Briscoe J., Singh, B., Ramasubban, R., Bhatia, R., (1995). Contingent valuation and actual behavior: predicting connections to new water systems in the state of Kerala, India, *The World Bank Economic Review*, 9: 373–95.

Guimarães, M.H., Sousa, C., Garcia, T., Dentinho, T., Boski, T., (2011). The value of improved water quality in Guadiana estuary—a transborder application of contingent valuation methodology, *Letters in Spatial and Resource Science*, 4: 31–48.

Haab, T.C., and Kenneth E., McConnell, (2002). Valuing Environmental and Natural Resources. Edward Elgar: Cheltenham, UK.

Halkos, G.E., Matsiori, S., (2011). *Economic valuation of coastal zone quality Improvements*, MPRA Paper 35395, University Library of Munich, Germany.

Halkos, G.E., Jones N., (2012). Modelling the effect of social factors on improving biodiversity protection, *Ecological Economics*, 78: 90-99.

Halkos, G.E., Matsiori, S., (2012a). Determinants of willingness to pay for coastal zone quality improvement, *The Journal of Socio-Economics*, 41: 391–399.

Halkos, G.E., Matsiori, S., (2012b). Assessing the economic value of protecting artificial lakes, MPRA Paper 39557, University Library of Munich, Germany.

Halpern, B.S., Selkoe, K.A., Micheli, F., Kappel, C.V., (2007). Evaluating and ranking the vulnerability of global marine ecosystems to anthropogenic threats, *Conservation Biology*, 21: 1301–1315.

Halpern, Ben S., et al., (2008). Managing for cumulative impacts in ecosystem based management through ocean zoning, *Ocean and Coastal Management*, 51: 203-211.

Hanley, N., Shogren, J.F., White, B., (1997). *Environmental economics in theory and practice*. England: Macmillan.

Hanley, N., MacMillan, D., Wright, R.E., Bullock, C., Simpson, I., Parrison, D. & Crabtree, R., (1998). Contingent valuation versus choice experiments: estimating the benefits of environmentally sensitive areas in Scotland, *Journal of Agricultural Economics*, 49: 1–15.

Hanley, N., Kristrom B., (2002). *What's It Worth? Exploring Uncertainty Over Values using Contingent Valuation*. Discussion papers in Economics no. 19/02, Economics Dept., University of Glasgow.

Hanley, N., Bell, D., Alvarez-Farizo, B., (2003). Valuing the benefits of coastal water quality improvements using contingent and real behaviour. *Environment and Resource Economics*, 24: 273-285.

Hanemann, W.M., (1984). Welfare Evaluations in Contingent Valuation Information with Discrete Responses, *American Journal of Agricultural Economics*, 66: 332 341.

Hanemann, W.M., (1994). Valuing the Environ-ment Through Contingent Valuation, *Journal of Economic Perspectives*, 8(4): 19-43.

Hanemann, M., Loomis, L., Kanninen, B., (1991). Statistical Efficiency of Double-Bounded Dichotomous Choice Contingent Valuation, *American Journal of Agricultural Economics*, 73: 1255-1263.

Harrison, G.W., (2006). Experimental evidence on alternative environmental valuation methods, *Environmental and Resource Economics*, 34(1): 125–62.

Hayes, K.M., Tyrrell, T.J., Anderson, G., (1992). Estimating the benefits of water quality improvements in the upper Narragansett Bay, *Marine Resource Economics*, 7: 75–85.

Heckman, J.J., (1979). Sample Selection Bias as a Specification Error, *Econometrica*, 47: 153-161.

Henry, C., (1974a). Option values in the economics of irreplaceable assets, *Review of Economic Studies*, 41: 89-104.

Henry, C., (1974b). Investment decision under uncertainty, the irreversible effect, *American Economic Review*, 64: 1006-1012.

Herrero, L.C., Sanz, J.A., Devesa, M., (2011). Measuring the economic value and social viability of a cultural festival as a tourism prototype, *Tourism Economics*, 17(3): 639–653.

Horowitz, J.L., (1992). A smoothed maximum score estimator for the binary response model, *Econometrica*, 60: 505-531.

Huang, J.C., Nychka, D.W., Smith, K.V., (2008). Semi-parametric discrete choice measures of willingness to pay, *Economics Letters*, 101: 91–94.

Jones, N., Sophoulis C.M., Malesios C., (2008). Economic valuation of coastal water quality and protest responses: A case study in Mitilini, Greece, *Journal of Socio-Economics*, 37: 2478–2491.

Jørgensen, B.S., Syme, G., Bishop, B., Nancarrow, B., (1999). Protest responses in contingent valuation, *Environmental and Resource Economics*, 14: 131–150.

Jørgensen, S.L., Olsen, S.B., Ladenburg, J., Martinsen, L., Svenningsen, S.R., Hasler, B., (2012). Spatially induced disparities in users' and non-users' WTP for water quality improvements—Testing the effect of multiple substitutes and distance decay, *Ecological Economics*. Available on http://dx.doi.org/10.1016/j.ecolecon.2012.07.015

Kahneman, D., Knetsch, J.L., (1992). Valuing public goods: The purchase of moral satisfaction, *Journal of Environmental Economics and Management*, 22: 57–70.

Katsanevakis, S., Stelzenmüller, V., South A., Sørensen T.K., Jones, P.J.S., Kerr, S., et al., (2011). Ecosystem-based marine spatial management: review of concepts, policies, tools and critical issues, *Ocean and Coastal Management*, 54:807–20

Kealy, M.J, Montgomery, M., Dovidio, J.F., (1990). Reliability and predictive validity of contingent values: does the nature of the good matter?, *Journal of Environmental Economics and Management*, 19: 244–63.

Khan, S., (2012). Distribution Free Estimation of Heteroskedastic Binary Choice Models using Probit Criterion Functions, *Journal of Econometrics*, Forthcoming.

Kim, S.S., Wong, K.F., Cho, M., (2007). Assessing the economic value of a world heritage site and willingness-to-pay determinants: a case of Changdeok Palace, *Tourism Management*, 28(1): 317-322.

Kontogianni, A., Langford, I.H., Papandreou, A., Skourtos, M.S., (2003). Social preferences for improving water quality: an economic analysis of benefits from wastewater treatment. *Water Ressources Management*, 17: 317–336.

Kountouri, P., Kountouris, Y., Remoundou, K., (2009). Valuing a wing farm construction: a contingent valuation study in Greece, *Energy Policy*, 37: 1939-1944.

Krinsky, I., and Robb, A.L., (1986). On approximating the statistical properties of elasticities, *Review of Economics and Statistics*, 68: 715-719.

Kriström B., (1990). A non parametric approach to the estimation of welfare measures in discrete response valuation, *Land Economics*, 66: 135-139.

Krutilla, J., (1967). Conservation reconsidered, *American Economic Review*, 57: 777–786.

Landry, C.E., Liu, H., (2009). A semi-parametric estimator for revealed and stated preference application to recreational beach visitation, *Journal of Environmental Economics and Management*, 57:205–218.

Limburg, K.E., O'Neil, R.V., Costanza, R., Farber, S., (2002). Complex systems and valuation, *Ecological Economics*, 41: 409–420.

Lipton, D., (2003). *The Value of Improved Water Quality to Chesapeake Bay Boaters*. Working Paper WP 03-16, Department of Agricultural and Resource Economics, University of Maryland, College Park.

List, J.A., Gallet, C.A., (2001). What experimental protocol influence disparities between actual and hypothetical stated values?, *Environmental and Resource Economics*, 20: 241–254.

Lockwood, M., Tracey, P., Klomp, N., (1996). Analysing Conflict between Cultural Heritage and Nature Conservation in the Australian Alps: A CVM Approach, *Journal of Environmental Planning and Management*, 39: 357–370.

Loomis, J.B., (1988). Broadening the concept and measurement of existence value, *Northeastern Journal of Agricultural and Resource Economics*, 17: 23–29.

Loomis, J., (1990). Comparative Reliability of the Dichotomous Choice and Open-Ended Contingent Valuation Techniques, *Journal of Environmental Economics and Management*, 18: 78-85.

Louviere, J., Woodworth G., (1983). Design and Analysis of Simulated Consumer Choice or Allocation Experiments: An Approach Based on Aggregate Data, *Journal* of Marketing Research, 20: 350-367. Luisetti, T., Turner, R.K., Hadley, D., Morse-Jones, S., (2010). *Coastal and Marine Ecosystem Services Valuation for Policy and Management*, CSERGE Working Paper EDM 10-04.

Madani, S., Ahmadian, M., KhaliliAraghi, M., Rahbar, F., (2012). Estimating Total Economic Value of Coral Reefs of Kish Island (Persian Gulf), *International Journal of Environmental Research*, 6(1):51-60.

Madden, Christopher J., Dennis H. Grossman, and Goodin, Kathleen L., (2005). Coastal and Marine Systems of North America: *Framework for an Ecological Classification Standard:Version II*. NatureServe, Arlington, Virginia.

Martinez, M.L., Intralawan, A., Vasquez, G., Pérez-Maqueo, O., Sutton, P., Landgrave, R., (2007). The coasts of our world: ecological, economic and social importance, *Ecological Economics*, 63: 254–272.

McFadden, D., (1974). *Conditional logit analysis of qualitative choice behavior*. In Frontiers in Econometrics, ed. P. Zarembka, New York: Academic Press, 105.42.

Md Nor, N.G., Mohd Yusoff, M. F. (2003). Value of Life of Malaysian Motorists: Estimates from a Nationwide Survey, *Journal of the Eastern Asia Society for Transportation Studies*, 5: 3031-3037.

Millennium Ecosystem Assessment (2005). *Millennium Ecosystem Assessment Synthesis Report*. Island Press, Washington, DC.

Mitchell, R.C., Carson, R.T., (1989). Using Surveys to Value Public Goods: The Contingent Valuation Method. Resources for the Future, Washington, DC.

Murphy, J.J., Allen, P.G., Stevens, T.H., Weatherhead, D., (2005). A meta-analysis of hypothetical bias in stated preference valuation, *Environmental and Resource Economics* 30: 313-25.

Nahman, A. and Rigby, D., (2008). Valuing Blue Flag Status and Estuarine Water Quality in Margate, South Africa, *South African Journal of Economics*, 76(4): 721-737.

Norton, B.G., (2000). Biodiversity and environmental values: In search of a universal earth ethic, *Biodiversity and Conservation*, 9:1029-1044.

Noonan, D.S., (2002). Contingent Valuation Studies in the Arts and Culture: An Annotated Bibliography. Working Paper No. 11. Cultural Policy Center, University of Chicago.

Nunes, P.A.L.D., van den Bergh, J.C.J.M., (2001). Economic valuation of biodiversity: sense or nonsense?, *Ecological Economics*, 39:203–222.

Nunes, P.A.L.D., Ding, H. and Markandya A., (2009). *The Economic Valuation of Marine Ecosystems*. FEEM Working Paper, No. 68.

O'Brien, B., Gafni, A., (1996). When do the "Dollars" make sense? Toward a conceptual framework for contingent valuation studies in health care, *Medical Decision Making*, 16: 288-299.

O'Neill, J., (1992). The varieties of intrinsic value, *The Monist*, 75(2): 119-137.

Organtzi, M., Mallios Z., and Latinopoulos, P., (2009). *Double bounded contingent valuation of quality improvement in a coastal environment*, In CEST2009: A-1030-1037.

Osorio H.M., Mittelhammer, R., (2012). *Estimating willingness to pay for recreation Site Attributes Using Information-Theoretic Methods*. Department of Agricultural and Resource Economics. Washington State University.

O'Shea, E., Gannon, B. and Kennelly, B., (2008). Eliciting Preferences for Resource Allocation in Mental Health Care in Ireland, *Health Policy*, 88: 359-370.

Park, T., Loomis, J.B., Creel, M., (1991). Confidence intervals for evaluating benefits estimates from dichotomous choice contingent valuation studies, *Land Economics*, 67(1): 64-73.

Pearce, D., (2002). *Handbook of Biodiversity Valuation: A Guide for Policy Makers*, Organisation for Economic Co-operation and Development, Paris, 156.

Pearce, D., Atkinson, G., Mourato, S., (2006). Cost-Benefit Analysis and the Environment, Recent Developments, OECD Publishing, Paris.

Pendleton, L., Atiyah, P., Moorthy, A., (2007). Is the non market literature adequate to support coastal and marine management?, *Ocean and Coastal Management*, 50: 363–78.

Petrolia, D.R., Kim T.G, (2009). What are barrier islands Worth? Estimates of Willingness to Pay for Restoration, *Marine Resource Economics*, 24: 131-146.

Poe, G.L., Clark, J.E., Rondeau, D., Schulze, W.D., (2002). Provision point mechanisms and field validity tests of contingent valuation. *Environmental and Resource Economics*, 23:105–131.

Ramajo-Hernandez, J., Saz-Salazar, S., (2012). Estimating the non-market benefits of water quality improvement for a case study in Spain: A contingent valuation approach, *Environmental Science and Policy*, 22: 47-59.

Remoundou, K., Koundouri, P., Kontogianni, A., Nunes, P.A.L.D., Skourtos M., (2009). Valuation of natural marine ecosystems: an economic perspective. *Environmental Science and Policy*, 12: 1040-1051.

Ready, R.C., Hu, D., (1995). Statistical approaches to the fat tail problem for dichotomous choice contingent valuation, *Land Economics*, 71(4): 491-499.

Rothe, C., (2009). Semiparametric Estimation of Binary Response Models With Endogenous Regressors, *Journal of Econometrics*, 153: 51-64.

Ryan, M., (2004). A comparison of stated preference methods for estimating monetary values, *Health Economics*, 13: 291-296.

Ryan, M., Watson, V., (2008). Comparing welfare estimates from payment card contingent valuation and discrete choice experiments, *Health Economics*, 18(4):389–401.

Salm, R.V., Clark, J., Siirila E., (2000). *Marine and Coastal Protected Areas: A guide for planners and managers*. IUCN. Washington DC. Xxi + 371pp.

Samuelson, P., (1954). The pure theory of public expenditure, *Review of Economics and Statistics*, 36: 332–338.

Schulze, W.D., d'Arge R.C., Brookshire D.S., (1981). Valuing environmental commodities: some recent experiments, *Land Economics*, 57:151–169.

Silberman, J., Klock, M., (1989). The behavior of respondents in contingent valuation: evidence on starting bids, *Journal of Behavioral Economics*, 18: 51–60.

Skourtos, M., Kontogianni, A., Georgiou, S., Turner, R.K., (2005). Valuing Coastal Systems. In R. K. Turner, W. Salomons & J. Vermaat (Eds.), *Managing European Coasts: Past, Present and Future* (pp. 119-136): Springer Verlag.

Smith, R.D., (2003). Construction of the contingent valuation market in health care: a critical assessment, *Health Economics*, 12: 609-628.

Stone, K., Bhat, M., Bhatta, R., Mathews, A., (2008). Factors influencing community participation in mangroves restoration: A contingent valuation analysis, *Ocean and Coastal Management*, 51: 476-484.

Strazzera, E., Genius, M., Scarpa, R., Hutchinson, G., (2003). The effect of protest votes on the estimates of WTP for use values of recreational sites, *Environmental and Resource Economics*, 25: 461–476.

Swanson, J., (2002). *Economic Valuation with Stated Preference Techniques*, Edward Elgar, Northampton, MA.

Tobin, J., (1958) Estimation of Relationships for Limited Dependent Variables, *Econometrica*, 26: 24- 36.

Turnbull, B.W., (1976). The empirical distribution function with arbitrarily grouped, censored and truncated data, *Journal of the Royal Statistical Society, Series B, Methodological*, 38: 290–295.

Turner, R.K., van den Bergh, C.J.M., Soderqvist, T., Barendregt, A., van der Straaten, J., Maltby, E., van Ierland, E.C., (2000). Ecological– economic analysis of wetlands: scientific integration for management and policy, *Ecological Economics*, 35: 7–23.

UNEP, (2006). Marine and coastal ecosystems and human wellbeing: A synthesis report based on the findings of the Millennium Ecosystem Assessment. UNEP. 76pp.

Van Biervliet K., Le Roy N., Nunes P. A. L. D., (2006). *An Accidental Oil Spill Along the Belgian Coast: Result from a CV Study*, Working Paper, FEEM No 41.

Venkatachalam, L., (2004). The contingent valuation method: a review, *Environmental Impact Assessment Review*, 24: 89–124.

Vilkka, L., (1997). *The intrinsic value of nature*. Amsterdam. The Netherlands: Rodopi.

Vossler, C.A., Kerkvliet, J., Polasky, S., Gainutdinova, O., (2003). Externally validating contingent valuation: an open-space survey and referendum in Corvallis, Oregon, *Journal of Economic Behaviour and Organization* 51: 261–277.

Wang, H., Laplante, B., Wu, X., Meisner C., (2004). *Estimating Willingness-to-Pay with Random Valuation Models: An Application to Lake Sevan, Armenia*, World Bank Policy Research, Working Paper No 3367.

Whitehead, J.C., Blomquist, G.C., Hoban, T.J., Clifford, W.B. Clif, (1995). Assessing the validity and reliability of contingent values: a comparison of on-site users, off-site users, and non-users, *Journal of Environmental Economics and Management*, 29: 238–251.

Whittington, D., Smith, V.K., Okorafor, A., Okore, A., Jin Liu, L., McPhail, A., (1992). Giving respondents time to think in contingent valuation studies: a developing country application, *Journal of Environmental Economics and Management*, 22: 205–225.

Winter, C., Lockwood, M., (2005). A model for measuring natural area values and park preferences, *Environmental Conservation*, 32: 270–278.

Winter, C., (2007). The intrinsic, instrumental and spiritual values of natural area visitors and the general public: A comparative study, *Journal of Sustainable Tourism*, 15: 599-614.